Author's response to reviews

Title: Is excretory urogram mandatory in patients with up to 20 mm renal and ureteric stones treated by extra corporeal shock wave lithotripsy?

Authors:

Dr M Hammad Ather (hammad.ather@aku.edu)
Nuzhat Faruqui (nuzhat.faruqui@aku.edu)
Sobia Akhtar (sobiaakter@hotmail.com)
M Nasir Sulaiman (nasir.sulaiman@aku.edu)

Version: 4 Date: 10 Mar 2004

Dear editor,

Following changes have been made in the manuscript upon your request.
1. Affiliations- added for each author according to the standard format.
2. Abstract truncated. Second sentence form background line 2-3 and last sentence from the methods section removed.
3. Section headings- Unnecessary capital removed
4. references placed in [], rather than superscript.
6. References- written in the standard pubmed format
7. Tables- Borders have visible black lines now
8. Figures Cropped
9. Figure legends removed from figure.

Kind regards
M H Ather